
Jaroslav Zalesak, Institute of Physics, Prague/FNAL 
  

Sep 22, 2014 

NOvA Experiment Status 
All Experimenter’s Meeting 



FarDet: Data-Taking 
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Partition 1           Sep 8, 2014           Partition 2  

Partition 1 (status last week) 
 diblocks 6 – 14  
 Running ~99.5% active,  
 ALL COOLED, HV at full gain. 

Current status - Retrofitting: 
 APD removed from both DiB-{01-02}-side and 
DiB-{03,04,05}-top,  
 Continues on DiB-05-side (4 APDs). 
 Next step back to DiB-01-top  
 Using already retrofitted DiB-04-top (6 dcms) 
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FarDet – recent status: 
 Physics partition contains 9 DiBs in one stack (06-14),  
 Cosmics, mutli-point readout (4 points of the a hit pulse 

instead of one, switched on 9/11 to make better timing at 
FarDet. 

 Powered down for ~20 hours due to the network outage. The 
network rack had to be exchange to accommodate more equipments. 

 Maintenance of HVAC systems at the same time.  
 Recovered in couple of hours without (almost) any problems! 
 Retrofitting continues and reached northernmost part of detector 

(DiB-05), After it finishes go back to DiB-01,02-top. 
 We have already retrofitted one part, which got new APDs (with good 

fittings) and now running as cold with full HV (only 3/384 ch bad) 

Operations I 



 NearDet:  
 Powered down three times for weekends due to the planned 

outages at Fermilab, twice ACNet was not available (no control 
over the detector), and last weekend the planned power 
outage occurred (the underground was affected). 

 For the first time we faced problem with the NearDet chiller 
system. It showed up that one water valve (at the heater 
exchanger) was mounted in bad direction. 

 Second time everything went ok. 
 Last weekend even control/safety systems were down, just dry 

gas system backup was available, detector is now up and 
running. 
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Operations II 
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ROC-west: 
 Announced the opening of Fermi operation center last week 
  not fully completed (door(s) missing,..)  
 Anyway prepared to move in! 

 Preliminary date 
set to Tue, Sep 30. 

 Need to make sure we  
have the 6 NOvA  
machines correctly set up 

Almost done. 

Operations III  
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